Author's response to reviews

Title: Air pollution exposure during pregnancy and reduced birth size: a prospective birth cohort study in Valencia, Spain.

Authors:

- Ferran Ballester (ballester_fer@gva.es)
- Marisa Estarlich (estarlich_mar@gva.es)
- Carmen Iñiguez (inyiguez_car@gva.es)
- Sabrina Llop (llop_sab@gva.es)
- Rosa Ramón (ramon_robson@gva.es)
- Ana Esplugues (esplugues_ana@gva.es)
- Marina Lacasaña (marina.lacasana.easp@juntadeandalucia.es)
- Marisa Rebagliato (rebagli@umh.es)

Version: 4 Date: 17 January 2010

Author's response to reviews:

MS: 1903277862317924

Air pollution exposure during pregnancy and reduced birth size in a mother and child cohort in Valencia, Spain.

Ferran Ballester, Marisa Estarlich, Carmen Iñiguez, Sabrina Llop, Rosa Ramón, Ana Esplugues, Marina Lacasaña and Marisa Rebagliato

Following the comments from the Independent Editorial Production Team of 15 January 2010 I am sending the revised submission of the referenced manuscript.

Ferran Ballester